Author’s response to reviews

Title: Testing an Active Intervention to Deter Researchers’ Use of Questionable Research Practices

Authors:

Samuel Bruton (samuel.bruton@usm.edu)
Mitch Brown (MitchellBrown@usm.edu)
Donald Sacco (Donald.Sacco@usm.edu)
Ralph Didlake (rdidlake@umc.edu)

Version: 1 Date: 02 Aug 2019

Author’s response to reviews:

I believe I have revised the manuscript in accord with your correspondence from Aug. 2, 2019. Specifically, on the assumption that the problem with the “figure” is that it would be more appropriately described as a table, and in light of the “Preparing Tables” section of the “Preparing your Manuscript” guidance, I have formatted it as a table object and placed it in the appropriate location within the manuscript.